Author’s response to reviews

Title: Factors associated with mental health consultation in South Korea

Authors:

Jieun Jang (jieun99@yuhs.ac)
Sang Ah Lee (ivory0817@yuhs.ac)
Woorim Kim (woorimk@yuhs.ac)
Young Choi (CHOIYOUNG223@yuhs.ac)
Eun-Cheol Park (ecpark@yuhs.ac)

Version: 3 Date: 02 Jan 2018

Author’s response to reviews:

Third Revision Note for MS ID: BPSY-D-17-00148R2

Factors associated with mental health consultation in South Korea

(BPSY-D-17-00148R2)

We thank you for giving us the opportunity to revise our paper. In revising our paper, we have carefully considered the editors’ comments and suggestions, and have done our best to incorporate them accordingly. We appreciate the constructive feedback on our original submission. After addressing the issues raised, we feel that the quality of the paper has improved and we hope you agree. Our responses to each comment are listed below. Overall, we modified what we could edit. We have attached revision notes. Once again, thank you for the valuable and helpful comments.
Response to Technical Comments

Editor comments:

I apologize if my comments on the changes to the figures was unclear. Thank you for making the correction by adding the figure legend to the body. Please remove the figure legends from the figure files so they only have the graphs. I hope that is clear. BR.

Response:

We thank you for giving us the opportunity to revise our paper. I added a figure title in the main body. Also, I remove the figure legends from the figure files so they only have the graph (Manuscript 5 file: page 12, line 10, line 17). If I do not understand editor comments or if it does not fit well of editor comment, please editor could fix the figure and manuscript.